Carers and Families

A follow-up study of the "Lighthouse" mentalization-based parenting program: Mentalization as a mediator of change

Clinical Child Psychology and Psychiatry 2024, Vol. 29(3) 867–881 © The Author(s) 2023 © ① ③

Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/13591045231220965 journals.sagepub.com/home/ccp

S Sage

Lina Gervinskaitė-Paulaitienė¹, Matthew Ruggiero², Svenja Taubner³, Jana Volkert⁴ and Rasa Barkauskienė¹

Institute of Psychology, Vilnius University, Lithuania

²Lifespan Psychology Centre, Australia

³Institute for Psychosocial Prevention, University of Heidelberg, Germany

⁴Clinic for Psychosomatic Medicine and Psychotherapy, University of Ulm, Germany

Abstract

This paper reports follow-up findings for an Mentalization based treatment (MBT) parenting intervention delivered to a community mental health sample. Parents completed the 12-week version of the Lighthouse Parenting Program (LPP) and were evaluated on parenting practices, parent-child relationships, parental mental health indicators, and child problem behaviour levels. We evaluated the extent to which improvements in mentalizing at follow-up mediated changes in parenting, parental adjustment, mental health, and child outcomes. Results included a reduction in parental coercive behaviours and child problems, improved parent-child relationship, and better parental psychological adjustment and mental health. Improvement in self-focused mentalizing were observed. Self-focused mentalizing mediated the changes in most outcomes from baseline to 3-month follow-up. These results provide strong preliminary evidence that the LPP improves parent and child outcomes.

Plain language summary: This paper presents the findings from a follow-up study that focused on a Mentalization based treatment (MBT) parenting intervention called the Lighthouse Parenting Program (LPP). This program is designed to enhance parents' ability to understand and respond sensitively to their own mental states (self-focused mentalizing) and to those of their children (childfocused mentalizing) and to strengthen parent-child relationships. We invited parents who had been referred for mental health services or whose children had been referred for mental health services to participate in the 12-week version of the LPP. Our study included 101 parents (82 mothers and 19 fathers). Participants were asked to complete study questionnaires about their parenting

Corresponding author:

Lina Gervinskaitė-Paulaitienė, Institute of Psychology, Vilnius University, Universiteto 9, Vilnius 01513, Lithuania. Email: lina.gervinskaite@fsf.vu.lt practices, mentalization, parent-child relationships, parental mental health indicators, and child problem behaviours before starting the program and 3-month after the program. We also examined whether improved mentalization predicted changes in parenting behaviors, parental wellbeing, and child problems. Results showed that parents were less likely to engage in coercive behaviors with children, their children exhibited fewer behavioral problems. Parent-child relationships improved, and parents reported enhanced emotional well-being and increased confidence in handling parenting tasks. There was also an improvement in self-focused mentalizing, i.e., how parents understood themselves, their thoughts, feelings and behaviors. The improved parental ability to understand their own mind played a significant role in most of the observed changes in parenting and child outcomes from before the program to three months after its completion. These results provide strong initial evidence that the LPP brings benefits to both parents and children.

Keywords

Parenting group, mentalization based treatment, parents, follow-up effects, parental mentalizing, child maltreatment prevention

Introduction

Parenting can be related to various positive developmental outcomes. By contrast, parenting behaviours can also increase a child's psychosocial vulnerability and in some instances cause acute and long-term harm (Toth & Manly, 2019). Child abuse, unresponsive caregiving, and neglect impose a heavy social burden, and can become perpetuated intergenerationally (Langevin et al., 2021). The most significant parent factors implicated in cases of maltreatment include parental mental health (Langevin et al., 2021), personality disorder traits (Florange & Herpertz, 2019), parenting stress (Pereira et al., 2012), parental self-efficacy (Michl-Petzing et al., 2019), relational factors (Langevin et al., 2021), as well as parental mentalization (Sharp & Fonagy, 2008). Even in non-clinical populations these factors appear to predict daily parenting behaviours (Michl-Petzing et al., 2019). Therapeutic and psychoeducational interventions such as parenting programs that target these factors can reduce risk of maltreatment and improve child developmental outcomes. Yet more work is required to understand the mechanisms by which such important changes are generated.

Parenting programs appear to be effective at reducing child maltreatment risk factors and recurrence (Altafim & Linhares, 2016; Branco et al., 2022). Follow-up investigations are important for distinguishing those interventions that produce lasting changes from those that do not. Effect sizes at follow-up tend to be small; indeed, for some parents, the behavioural changes apparent at post-treatment diminish or disappear over time, although it is unclear why this occurs (Backhaus et al., 2023). More research is needed to understand the ways in which intervention, when it works, affects parent-child psychological and relational processes.

Mentalizing capacity and parenting behaviour

Mentalization is the capacity to infer the mental states (emotions, thoughts, beliefs, needs, intentions in oneself and others) that motivate behaviour and to utilise this information to navigate social interaction (Fonagy & Bateman, 2007). This multi-faceted and fluctuating capacity plays an important role in parenting and child development (Camoirano, 2017; Sharp & Fonagy, 2008).

Mentalization also appears to scaffold parents and protect their experience of parenting under stress. For instance, mentalization predicts higher parental satisfaction in those parents who have themselves experienced disrupted attachment (Burkhart et al., 2017), it predicts positive parenting behaviours (Santelices & Cortés, 2022), and it reduces parental hostility when parents have poor emotion regulation (Gershy & Gray, 2020). Essentially, when parent and child internal states can be mentally represented, they can be converted from states of affective intensity into symbolic thought. This may be the cornerstone of emotional self-regulation and a range of consequent social skills.

In contrast, low parental mentalization predicts a range of vulnerabilities and problems for children and parents (Condino et al., 2022; Rostad & Whitaker, 2016). Mentalizing capacity distinguishes mothers whose attitudes and behaviours predict child abuse risk from those who show no significant indicators of risk (Hunter et al., 2022). Parents with a history of child maltreatment demonstrate hostility to mentalizing demands (Rosso, 2022). Synthesising the present state of this literature, Byrne et al. (2019) have suggested that child abuse can be understood as a consequence of dysfunctional or limited mentalizing capacity and the consequent risk that children's mental states and behaviours can be interpreted in distorted or inaccurate ways. Mentalizing may therefore be an important consideration in efforts to reduce the social impact of child abuse and neglect. Researchers can begin to distinguish whether successful parenting interventions are those that improve one or more facets of parental mentalization (thus reducing the likelihood that affective impulses become expressed as problematic behaviours). This is a directly testable hypothesis, which can be evaluated by looking at the ways in which mentalizing mediates any impact of intervention on improved parenting outcomes.

Follow-up effects in mentalization-based parenting treatments

Mentalization based treatments (MBT) for parents attempt to improve their capacity to understand and respond sensitively to their own mental states (self-focused mentalizing) and to those of their children (child-focused mentalizing). Outcome evaluations are published for several such interventions (for recent reviews, see: Lavender et al., 2023; Lo & Wong, 2022). MBT interventions for parents appear to improve parental mentalizing, mental health, and the quality of caregiving (Camoirano, 2017). Relatively few studies have reported follow-up data in relation to such changes and no consistent pattern of results is yet detectable in those that have. For instance, Ordway et al. (2014) reported that an MBT parenting treatment improved child behavioural outcomes that were maintained at follow-up. Mentalizing continued improving beyond the termination of treatment, yet these improvements did not appear to relate to child behaviour. Suchman et al. (2011) reported that changes in one specific aspect of parental mentalizing (self-focused) were maintained at follow-up for the same intervention. Later results for the same intervention in a community mental health sample indicated improvement in child-focused, but not self-focused mentalizing, at follow-up (Suchman et al., 2017).

Conflicting results are also apparent in relation to psychological outcomes for MBT Parenting interventions. Several studies have reported that intervention groups cannot be differentiated from control groups or baseline measurements on depression and psychopathology measures at follow-up (Ordway et al., 2014; Suchman et al., 2011, 2017). Yet others demonstrate impressive changes in these factors (Salo et al., 2019; Suchman et al., 2017). There are many possible explanations for these observations, including sample characteristics and the range of ways in which mentalization has been operationalised and measured. No study has evaluated whether specific facets of mentalizing mediate these psychological outcomes.

Lastly, child outcomes have been reported in very few follow-up studies. Early evidence suggests that MBT parenting interventions generate improvements for children (Enav et al., 2019; Midgley et al., 2019), although much more is needed in this regard.

Current study

Early evidence suggests that MBT for parents improves parenting behaviours, family relationships, and mental health correlates (each of which are implicated in child abuse risk and parenting behaviour). However, these effects have mainly been observed at post-treatment and follow-up evaluations are rare. In addition, it is not clear which facets of mentalizing (if any) are mediators of change when a parenting intervention produces improvements in parenting variables. The present study directly addressed these gaps by evaluating parenting variables 3-month after the conclusion of an MBT Parenting intervention. Post-treatment effects for this intervention have been reported elsewhere (Gervinskaitė-Paulaitienė et al., 2023). In the current study we evaluated the extent to which improvement in mentalizing capacity mediated changes in parenting practices, parent-child relationships, family relationships, parental mental health indicators, and child problem behaviour levels. Three facets of mentalizing were tested: (i) self-focused, (ii) other-focused, and (iii) motivation to mentalize. We hypothesized that at follow-up, all outcomes would show improvements relative to baseline measurements. We hypothesised that all three facets of mentalizing would be higher at follow-up relative to baseline measurements. Lastly, we hypothesised that these changes in parenting, psychological adjustment and mental health, family functioning, and child problems would be mediated by improvements in each of the three facets of mentalizing.

Method

Design

This was a non-controlled one group pre-posttest study with 3-month follow-up. The study was approved by the Institutional Ethics Committee. Informed consent from all participants was obtained prior to their participation. Outcome measurements were taken at three time points: (i) before intervention (baseline); (ii) after intervention ended (post-test); and (iii) at 3-month follow-up. Participants completed all questionnaires online, accessing them via separate links. The post-test analysis of this study has been published elsewhere (Gervinskaité-Paulaitiené et al., 2023).

In this paper we focus on the analysis of the 3-month follow-up. This study was pre-registered: https://doi.org/10.17605/OSF.IO/HX7VB

Participants

This study was implemented in five community mental health institutions who provide treatment for families and children. The sample consisted of 101 parents (82 mothers and 19 fathers) living in two cities in Lithuania (age range = 24–60; M = 39.1, SD = 5.74). Most parents were married/ cohabitating (78%), 10% were divorced, 8% were single and 4% other. Most families had two children (54%), 23% had one child, and 23% had three or more children. Most of the parents had higher education (89%), 9% completed secondary education, and 1% had not completed school. Most parents were employed (80%), 13% were unemployed, and 7% were mothers on parental leave. Most parents had children with prior referrals to mental health or psychoeducational services because of a child's problems (73%) and 45% had children who were currently receiving services.

Many parents had their own prior experiences with a mental health service (73%) and 44% were currently receiving services. Parents were recruited in collaboration with group facilitators and other specialists. Invitations to participate in the study were also distributed through online groups and mailing lists related to parenting.

Inclusion criteria were: (i) have at least one child, aged 12 or less; (ii) parents themselves had been referred or self-referred for mental health services due to mental health, child rearing or family problems, and/or have a child who had been referred for mental health or psychoeducational services due to their emotional, behavioural, or developmental problems; (iii) speak Lithuanian; and (iv) agree to participate in the study. Parents were excluded from the study if: (i) their referral required inpatient treatment; (ii) their referral indicated incompatibility with group treatment; (iii) they experienced significant intellectual disability; or (iv) they were in another parenting program.

Measures

Parenting and family adjustment scales (Sanders et al., 2014). This scale assesses parenting practices and parent/family functioning. It contains 30 items and has 7 subscales. Higher scores denote higher levels of dysfunction. In this study, Cronbach's α was good for all subscales except one, which was unacceptable and thus excluded from analyses (Coercive Parenting = .71; Positive Encouragement = .72; Parent–Child Relationship = .85; Parental Adjustment = .80; Family Relationships = .78; Parental Teamwork = .79; Parental Consistency = .38).

The parental stress scale (Berry & Jones, 1995). This scale evaluates stress related to child rearing, capturing demands and rewards of parenting. The 18 items sum to produce a total stress score, which in this study had good reliability ($\alpha = .89$).

Brief parental self efficacy scale (Woolgar et al., 2023). This 5-item scale measures how confident parents are that they can manage the tasks and demands of parenting. It produces a single score, which in this study had good reliability ($\alpha = .81$).

Mentalization scale (Dimitrijević et al., 2018). This 28-item scale evaluate three aspects of mentalization: self-related (MentS-S), other-related (MentS-O), and motivation to mentalize (MentS-M). Higher subscale scores denote a higher capacity for mentalization. In this study, reliability ranged from acceptable to good ($\alpha_{Total} = .86$; $\alpha_{MentS-S} = .69$; $\alpha_{MentS-O} = .74$; $\alpha_{MentS-M} = .79$).

Patient health Questionnaire-9 (Kroenke et al., 2001). This 9-item scale screens respondents for depressive symptoms. It produces a single score, which in this study had good reliability ($\alpha = .85$).

The Generalized Anxiety Disorder scale-7 (Spitzer et al., 2006). This 7-item scale screens respondents for generalized anxiety disorder symptoms. It produces a single score, which in this study had excellent reliability ($\alpha = .91$).

The Level of Personality Functioning Scale-Brief Form 2.0 (Weekers et al., 2019). This 12-item scale assesses personality functioning, with higher scores indicating more impaired personality functioning. This scale had good reliability in this study ($\alpha = .83$).

Strengths and Difficulties Questionnaire (Goodman, 1997). This scale asks parents to rate their child's emotional and behavioural problems. We used a total problem score, which is obtained by summing

20 items that measure hyperactivity, emotional, conduct problems, and problems with peers. If parents had more than one child, they were asked to fill out the SDQ for whichever child experienced the most difficulties at that time.

Intervention

The Lighthouse Parenting Program (LPP) is a manualised MBT Parenting intervention originally developed for implementation with parents who have harmed or neglected their children (Byrne et al., 2019). It combines psychoeducational and psychotherapeutic interventions to improve parental mentalizing and strengthen parent-child relationships. The full program is 20-week long and comprises weekly group and fortnightly individual therapy sessions. We selected the 12-week adaptation of the program for this study (Byrne & Ruggiero, 2018) since it was developed for use with parents in community mental health settings. This version does not include individual therapy (group only). The program manual details the tasks, content, and recommended activities for each of the 12 modules. The aim of the program is to increase parental mentalizing through experiential activities, discussions of educational material, and by encouraging and modelling mentalizing in group sessions. MBT individual and group skills are employed by the facilitators throughout the program. The LPP uses metaphors (such as lighthouse, safe harbour, etc.) to convey attachment and mentalization concepts to parents.

All parents attended an individual intake session with one of the group facilitators, to prepare for entry into the program. Each session was 2 hours duration with a short break. There were nine groups running in five locations with 7–15 participants per group. Two facilitators led each group. All facilitators were qualified psychologists who received LPP training from the program authors (G.B. and M.R.). Facilitators attended weekly group supervision delivered by one program author (G.B.) and a second accredited LPP supervisor.

Data analysis

All analyses were performed using IBM SPSS Statistics (Version 27). To evaluate group differences, we used Mann-Whitney for outcome measures and Chi-square for demographic characteristics (comparing completers with drop-outs; comparing those who participated in the follow-up assessment and those who did not). A paired samples *t* test was performed to test the differences between outcome scores at baseline and follow-up. Cohen's d was calculated to interpret effect sizes. An alpha level of p < .05 was used for all analyses. Mediation hypotheses were tested using Mediation and Moderation Analysis for Repeated Measures Designs MEMORE (Montoya & Hayes, 2017). MEMORE is a SPSS macro available from the author's website (https://www.akmontoya.com/spss-and-sas-macros). MEMORE estimates total, direct, and indirect effects in a two-condition within-participant design. In this mediation analysis the independent variable was "intervention". In the analysis, the changes in dependent variables and mediators from baseline to follow-up were calculated. The significance of indirect effects was estimated using the bootstrap method of inference (with 5000 runs), providing 95% confidence intervals (CI) for an indirect effect. An indirect effect is considered statistically significant if the confidence interval excludes zero (Montoya & Hayes, 2017).

Results

From all parents (N = 101) who started the program, 89 (88%) parents finished it. From completers, 75 participated in follow-up (see Figure 1). Comparisons of completers (n = 89) and drop-outs (n = 12) showed no differences in outcomes, except for employment status (more employed parents finished the program and more mothers on maternity leave dropped out). Further, we compared parents who completed the program and the follow-up assessment (n = 75) with parents who completed the program but did not participate in follow-up (n = 14). They differed on MentS at baseline (Mann-Whitney U = 238.0, p = .010), parents who did not participate at follow-up reported higher MentS scores (mean rank = 60.67) compared to parents who did participate (mean rank = 40.72).

The results of comparisons of means for all variables at baseline to 3-month follow-up are presented in Table 1. There was a significant decrease in coercive parenting (t = 6.269, p < .001), parent-child relationship (t = 2.757, p = .007) and parental emotional adjustment (t = 5.448, p < .001), parental self-efficacy (t = -2.821 p = .003) improved at follow-up. Anxiety (t = 5.484, p < .001) and depressive symptoms (t = 3.065, p = .003) were lower at follow-up. The level of personality functioning also improved (t = 2.784, p = .007). Only self-mentalization significantly improved (t = -2.639, p = .010). Parents reported significantly lower levels of child's problems at follow-up (t = 4.585, p < .001).

Improvement in self-related mentalization was significant, whilst changes in other-related mentalization and motivation to mentalize were negligible. As our goal was to examine the indirect effects of mentalization, we ran the series of the MEMORE analysis to test the mediational effect with self-related mentalization and parenting and child outcomes. Table 2 displays the output of this mediational analysis.

Several significant mediational effects were detected. Improvement in self-related mentalization mediated the effect (at follow-up) of intervention on: (i) reduction of coercive parenting practices; (ii) improvements in parental emotional adjustment, family relationships, parental teamwork, self-efficacy, and personality functioning; (iii) decreases in parental stress and depressive symptoms; and (iv) reduction of child behavioural problems. However, improvement in self-related mentalization



Figure 1. Study flow.

	Baseline		3 month-follow up					
	М	SD	М	SD	t	df	Þ	d
PAFAS								
Coercive parenting	6.36	2.42	4.50	2.16	6.269	71	<.001	.739
Encouragement	3.24	1.97	2.85	1.93	1.798	71	.076	.212
Parent-child relationships	3.55	2.95	2.82	2.92	2.757	72	.007	.323
Parental adjustment	7.58	2.91	5.79	2.86	5.448	70	<.001	.647
Family relationships	4.94	2.39	4.53	2.81	1.408	71	.163	.166
Parental teamwork	3.78	2.36	3.38	2.03	1.545	57	.128	.203
Parental stress	40.36	11.50	38.41	10.44	1.714	65	.091	.211
Self-efficacy	18.81	3.54	19.88	2.60	-2.82 I	71	.006	332
Anxiety symptoms	8.24	5.06	5.03	3.92	5.484	67	<.001	.665
Depressive symptoms	7.96	5.03	6.04	4.60	3.065	66	.003	.375
Level of personality functioning	24.97	7.21	22.87	7.45	2.784	66	.007	.340
MentS								
Self-related mentalization	28.38	4.96	29.84	4.95	-2.639	68	.010	318
Other-related mentalization	38.90	4.46	38.62	4.30	.652	67	.516	.079
Motivation to mentalize	41.06	4.76	41.01	5.03	.094	70	.925	.011
Total	108.56	11.75	109.87	11.54	-1.101	62	.275	—.139
SDQ. Child's total problems	16.52	5.42	13.45	5.91	4.585	65	<.001	.564

Table I. Comparisons of outcomes at pre-test and follow-up.

did not mediate the effect of intervention on changes in parent-child relationship and positive encouragement.

Discussion

In our study, follow-up observations largely supported hypothesis 1: (i) coercive behaviours had reduced; (ii) the parent-child relationship had improved; (iii) parents were more emotionally adjusted; (iv) parents had less mental health symptoms; (v) parenting self-efficacy improved; and (vi) child problems had reduced. Parents had improved in their ability to mentalize their own internal states (partial support for hypothesis 2), and this improvement in mentalizing generated the changes we observed in most outcomes from baseline to follow-up (partial support for hypothesis 3). These results provide strong preliminary evidence that the LPP improves parent and child outcomes by influencing at least one facet of parental mentalization (self-mentalizing).

We had expected all dimensions of mentalization to improve after LPP. However, only selffocused mentalization improved, whereas no change was observed in other-focused and motivation to mentalize. Previous studies of MBT parenting interventions have measured mentalization related to the parent-child relationship, observing improvements (Enav et al., 2019; Salo et al., 2019). In a small study of LPP with high-risk parents, mentalization did not improve (Byrne et al., 2019). The few follow-up studies of MBT parenting interventions appear to show that improvements in mentalization are maintained (Salo et al., 2019; Sleed et al., 2013). The change in self mentalizing that we observed in this study is consistent with Suchman et al. (2011), confirming that it is important to treat this as a multi-dimensional phenomenon and not a single outcome in intervention

Table 2. Mediational analy	/sis.
----------------------------	-------

		Model summary		Indirect effect		Bootstrapping 95% Cl	
		R2	F	Effect	BootSE	Lower	Upper
Mediator:	Self-related mentalization						
Outcomes	:						
PAFAS	Coercive parenting	.11	3.781*	25	.11	470	046
	Encouragement	.13	4.638*	17	.10	382	.007
	Parent-child relationships	.05	1.576	14	.11	362	.061
	Parental adjustment	.10	3.516*	29	.16	—.619	02 I
	Family relationships	.17	6.475**	33	.19	—. 761	02 l
	Parental teamwork	.16	4.725*	25	.12	—.529	048
Parental :	stress	.15	5.104**	-1.26	.60	-2.495	131
Self-effica	су	.10	3.634*	.30	.19	.027	.735
Anxiety symptoms		.03	.901	30	.32	-1.028	.277
Depressive symptoms		.14	4.898**	64	.33	-1.390	—.099
Level of personality functioning		.19	7.067**	-1.01	.46	-2.036	23I
Child's problems		.18	6.575**	86	.36	-1.643	252

*p < .05, **p < .01, ***p < .001

research. Fonagy and Luyten (2018) have suggested that the capacity to represent internal self-states pre-dates the capacity to distinguish and represent others' minds as separate entities. So it may be unsurprising that a short-term mentalizing intervention has the greatest effect on this facet of mentalizing. Our results suggest that self-focused mentalizing may be of particular importance in parenting interventions.

The results observed here have several alternative explanations. The increase in self-focused mentalizing might be linked to the fact that LPP invites parents to attend to, and reflect, on their own emotional experiences in their attachment relationships and in the group. Participants are mentalized 'live', whereas their children are not present to observe when mentalizing turns to their experiences. Furthermore, our chosen mentalizing measure uses generalised item wording and does not specify, for example, that attention be paid to the child who was in focus during the program when responding to items. It is possible that any changes in other-focused mentalizing and motivation to mentalize are specific (related to target children only) and not yet generalised after the 12-week intervention. Without the inclusion of a control/comparison group and an evaluation of these capacities in relation to a target child, it is not possible to comment further on this result.

Our results show reduction in coercive parenting, i.e., parents less frequently shouted at children, got angry with them, used physical discipline, and engaged in control through inducing guilt and shame. As these behaviours are associated with a risk of child maltreatment (Sanders et al., 2014), their decrease is important for maltreatment prevention. Our results are comparable with other child abuse prevention programs (Altafim & Linhares, 2016). Moreover, we distinguished that this decrease in coercive parenting was mediated by improved self-related mentalization; that is, as parents better understand their own mental states, their coercive behaviour reduced. This is consistent with Fonagy's (2003) suggestion that lower mentalization is related with aggressive behaviours and provides a context for the relationship between parental emotion dysregulation and hostile parenting (Gershy & Gray, 2020). Increased self-focused mentalizing might help parents to

regulate themselves better in emotionally charged interactions with the child and not resort to coercive behaviours.

There was an improvement in the parent-child relationship quality. Parents felt more reciprocal warmth and were more satisfied with the relationship. It is important to note that there were no significant differences in the relationship from pre-to post-test in this sample (Gervinskaitė-Paulaitienė et al., 2023). It might be that these changes need time to unfold, as seen elsewhere (Suchman et al., 2017). Although related to the intervention, this improvement was not generated by the increase in self-focused mentalizing.

Parental emotional adjustment, self-efficacy, and family functioning are important risk factors in child maltreatment (Langevin et al., 2021; Michl-Petzing et al., 2019; Pereira et al., 2012). We observed each of these factors improve at follow-up. Other risk factors for maltreatment (parenting stress, family relationships/parental teamwork) did not improve. However, in the mediational analysis, changes in all five outcomes were predicted by changes in self-focused mentalizing. This is a complex, although important finding. Parenting stress on its own, for instance, may be a poor measure of intervention success, since in some parents (such as those who utilise primarily avoidant attachment processes) it might be hoped that the capacity to be affected by the task of parenting increase during treatment, so that it can then be regulated in new ways. In this study we observed that improvements in self mentalizing predicted improvements in all of the important predictors of maltreatment, regardless of whether parents' experiences of stress and family relationships became more or less severe.

Depressive symptoms, anxiety symptoms, and personality functioning all improved at followup. This is consistent with the extensive literature documenting that MBT interventions improve affective and personality disorder symptoms (Malda-Castillo et al., 2019). The present study suggests that a relatively short MBT parenting intervention might have mental health benefits. Improvements in self-focused mentalizing predicted the change in depressive symptoms and personality functioning, but not anxiety. There is a close link between depression and attachment processes (Halstensen et al., 2021). As such, LPP interventions pay close attention to parents' historic and current attachment processes throughout the program (Byrne et al., 2019). Parents describe that having opportunities to reflect on their attachment experiences in a safe group environment are an important element of the program (Gervinskaité-Paulaitiené et al., 2023). Our observations that personality functioning also improved via increases in self-focused mentalizing align with the conceptual and functional closeness others have identified between these phenomena (Zettl et al., 2020).

Parents indicated that their children's behavioural problems had reduced at follow-up. These results concur with prior studies reporting improvements in child problems after MBT parenting interventions (Enav et al., 2019; Midgley et al., 2019). The mediational analysis suggests that improved self-focused mentalizing predicts decreases in child problems. It is important to note that parents themselves report their child's problems. It cannot be ruled out that parents' views of their children, changed and that this subsequently influenced the way in which they perceived their children's problems.

Limitations and future directions

Several design limitations warrant consideration. There was no control/comparison group used for this study. Future studies of LPP should build in a control/comparison condition to improve internal validity and enable causal conclusions to be drawn. A 3-month follow-up is relatively short and replication studies could use longer periods where feasible. We used self-report measures for

outcome variables, which can be prone to social desirability and faking-good response patterns. Narrative-based measures and behavioural observations are more time intensive to administer but might reduce these assessment risks. Live evaluations of mentalizing are extremely time intensive but offer a gold-standard in terms of the delineation of mentalizing capacities. Our results suggest that improvements in self-focused mentalizing are an appropriate clinical target for intervention studies focused on parenting groups or problems (e.g., for those struggling with parental adjustment, low self-efficacy or using coercive practices), but that improvements in this single facet may not be sufficient to increase positive encouragement or parent-child relationship quality.

Conclusion

This study provides evidence that the effectiveness of MBT parenting interventions is detectable at follow-up. It was the first parenting intervention study to evaluate the ways in which facets of mentalizing mediate the effect of intervention on a range of parent and child outcomes. We showed that the 12-week adaptation of LPP produced a range of important outcomes in a community mental health sample of parents at follow-up. Reduction of coercive parenting practices, parental stress, child behaviour problems and improvements in parental mental health, family relationships, parental teamwork, and self-efficacy were mediated by the improvements in self-focused mentalizing that the intervention generated in the sample. Risk factors associated with child maltreatment were positively impacted by the program.

Declaration of conflicting interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) disclosed receipt of the following financial support for the research, authorship, and/or publication of this article: This research was funded by the European Social Fund (project No 09.3.3-LMT-K-712-230152) under grant agreement with the Research Council of Lithuania (LMTLT). This study was pre-registered: https://doi.org/10.17605/OSF.IO/HX7VB

Ethical statement

Ethical approval

The study was approved by the Institutional Ethics Committee, Vilnius University (protocol No. 6/(1.13 E) 250000-KT-23).

ORCID iD

Lina Gervinskaitė-Paulaitienė 💿 https://orcid.org/0000-0003-1464-8396

References

Altafim, E. R. P., & Linhares, M. B. M. (2016). Universal violence and child maltreatment prevention programs for parents: A systematic review. *Psychosocial Intervention*, 25(1), 27–38. https://doi.org/10.1016/j.psi. 2015.10.003

- Backhaus, S., Leijten, P., Jochim, J., Melendez-Torres, G. J., & Gardner, F. (2023). Effects over time of parenting interventions to reduce physical and emotional violence against children: A systematic review and meta-analysis. *EClinicalMedicine*, 60, 102003. https://doi.org/10.1016/j.eclinm.2023.102003
- Berry, J. O., & Jones, W. H. (1995). The parental stress scale: Initial psychometric evidence. Journal of Social and Personal Relationships, 12(3), 463–472. https://doi.org/10.1177/0265407595123009
- Branco, M. S. S., Altafim, E. R. P., & Linhares, M. B. M. (2022). Universal intervention to strengthen parenting and prevent child maltreatment: Updated systematic review. *Trauma, Violence, & Abuse, 23*(5), 1658–1676. https://doi.org/10.1177/15248380211013131
- Burkhart, M. L., Borelli, J. L., Rasmussen, H. F., Brody, R., & Sbarra, D. A. (2017). Parental mentalizing as an indirect link between attachment anxiety and parenting satisfaction. *Journal of Family Psychology: JFP: Journal of the Division of Family Psychology of the American Psychological Association (Division 43)*, 31(2), 203–213. https://doi.org/10.1037/fam0000270
- Byrne, G., & Ruggiero, M. (2018). Lighthouse parenting. Mentalization based group programme. Facilitator guide (Twelve Week Edition). Unpublished.
- Byrne, G., Sleed, M., Midgley, N., Fearon, P., Mein, C., Bateman, A., & Fonagy, P. (2019). Lighthouse Parenting Programme: Description and pilot evaluation of mentalization-based treatment to address child maltreatment. *Clinical Child Psychology and Psychiatry*, 24(4), 680–693. https://doi.org/10.1177/ 1359104518807741
- Camoirano, A. (2017). Mentalizing makes parenting work: A review about parental reflective functioning and clinical interventions to improve it. *Frontiers in Psychology*, 8, 14. https://doi.org/10.3389/fpsyg.2017. 00014
- Condino, V., Giovanardi, G., Vagni, M., Lingiardi, V., Pajardi, D., & Colli, A. (2022). Attachment, trauma, and mentalization in intimate partner violence: A preliminary investigation. *Journal of Interpersonal Violence*, 37(11–12), NP9249–NP9276. https://doi.org/10.1177/0886260520980383
- Dimitrijević, A., Hanak, N., Altaras Dimitrijević, A., & Jolić Marjanović, Z. (2018). The mentalization scale (MentS): A self-report measure for the assessment of mentalizing capacity. *Journal of Personality* Assessment, 100(3), 268–280. https://doi.org/10.1080/00223891.2017.1310730
- Enav, Y., Erhard-Weiss, D., Kopelman, M., Samson, A. C., Mehta, S., Gross, J. J., & Hardan, A. Y. (2019). A non randomized mentalization intervention for parents of children with autism. *Autism Research: Official Journal of the International Society for Autism Research*, 12(7), 1077–1086. https://doi.org/10.1002/aur. 2108
- Florange, J. G., & Herpertz, S. C. (2019). Parenting in patients with borderline personality disorder, sequelae for the offspring and approaches to treatment and prevention. *Current Psychiatry Reports*, 21(2), 9. https:// doi.org/10.1007/s11920-019-0996-1
- Fonagy, P. (2003). Towards a developmental understanding of violence. *The British Journal of Psychiatry: The Journal of Mental Science*, 183(3), 190–192. https://doi.org/10.1192/bjp.183.3.190
- Fonagy, P., & Bateman, A. W. (2007). Mentalizing and borderline personality disorder. Journal of Mental Health, 16(1), 83–101. https://doi.org/10.1080/09638230601182045
- Fonagy, P., & Luyten, P. (2018). Attachment, mentalizing, and the self. In W. J. Livesley & R. Larstone (Eds.), Handbook of personality disorders: Theory, research, and treatment (pp. 123–140). The Guilford Press.
- Gershy, N., & Gray, S. A. O. (2020). Parental emotion regulation and mentalization in families of children with ADHD. *Journal of Attention Disorders*, 24(14), 2084–2099. https://doi.org/10.1177/1087054718762486
- Gervinskaitė-Paulaitienė, L., Byrne, G., & Barkauskienė, R. (2023). Mentalization-based parenting program for child maltreatment prevention: A pre–post study of 12-week lighthouse group program. *Children*, 10(6), 1047. Article 6. https://doi.org/10.3390/children10061047
- Goodman, R (1997). The Strengths and Difficulties Questionnaire: a research note. J Child Psychol Psychiatry, 0021-963038(5), 581–586. https://doi.org/10.1111/j.1469-7610.1997.tb01545.x. 9255702.

- Halstensen, K., Gjestad, R., Luyten, P., Wampold, B., Granqvist, P., Stålsett, G., & Johnson, S. U. (2021). Depression and mentalizing: A psychodynamic therapy process study. *Journal of Counseling Psychology*, 68(6), 705–718. https://doi.org/10.1037/cou0000544
- Hunter, M., Bailey, K., & Casey, J. (2022). Risk factors in abusive parent and child relationships. *Neonatal*, 3(1), 6. https://doi.org/10.35702/neo.10006
- Kroenke, K., Spitzer, R. L., & Williams, J. B. W. (2001). The PHQ-9: Validity of a brief depression severity measure. *Journal of General Internal Medicine*, 16(9), 606–613. https://doi.org/10.1046/j.1525-1497. 2001.016009606.x
- Langevin, R., Marshall, C., & Kingsland, E. (2021). Intergenerational cycles of maltreatment: A scoping review of psychosocial risk and protective factors. *Trauma, Violence, & Abuse, 22*(4), 672–688. https:// doi.org/10.1177/1524838019870917
- Lavender, S. R., Waters, C. S., & Hobson, C. W. (2023). The efficacy of group delivered mentalization-based parenting interventions: A systematic review of the literature. *Clinical Child Psychology and Psychiatry*, 28(2), 761–784. https://doi.org/10.1177/13591045221113392
- Lo, C. K. M., & Wong, S. Y. (2022). The effectiveness of parenting programs in regard to improving parental reflective functioning: A meta-analysis. *Attachment & Human Development*, 24(1), 76–92. https://doi.org/ 10.1080/14616734.2020.1844247
- Malda-Castillo, J., Browne, C., & Perez-Algorta, G. (2019). Mentalization-based treatment and its evidencebase status: A systematic literature review. *Psychology and psychotherapy*, 92(4), 465–498. https://doi. org/10.1111/papt.12195
- Michl-Petzing, L. C., Handley, E. D., Sturge-Apple, M., Cicchetti, D., & Toth, S. L. (2019). Re-examining the "cycle of abuse": Parenting determinants among previously maltreated, low-income mothers. *Journal of Family Psychology: JFP: Journal of the Division of Family Psychology of the American Psychological Association (Division 43)*, 33(6), 742–752. https://doi.org/10.1037/fam0000534
- Midgley, N., Cirasola, A., Austerberry, C., Ranzato, E., West, G., Martin, P., Redfern, S., Cotmore, R., & Park, T. (2019). Supporting foster carers to meet the needs of looked after children: A feasibility and pilot evaluation of the reflective fostering programme. *Developmental Child Welfare*, 1(1), 41–60. https://doi. org/10.1177/2516103218817550
- Montoya, A. K., & Hayes, A. F. (2017). Two-condition within-participant statistical mediation analysis: A path-analytic framework. *Psychological Methods*, 22(1), 6–27. https://doi.org/10.1037/met0000086
- Ordway, M. R., Sadler, L. S., Dixon, J., Close, N., Mayes, L., & Slade, A. (2014). Lasting effects of an interdisciplinary home visiting program on child behavior: Preliminary follow-up results of a randomized trial. *Journal of Pediatric Nursing*, 29(1), 3–13, https://doi.org/10.1016/j.pedn.2013.04.006
- Pereira, J., Vickers, K., Atkinson, L., Gonzalez, A., Wekerle, C., & Levitan, R. (2012). Parenting stress mediates between maternal maltreatment history and maternal sensitivity in a community sample. *Child Abuse & Neglect*, 36(5), 433–437. https://doi.org/10.1016/j.chiabu.2012.01.006
- Rosso, A. M. (2022). When parents fail to mind the child: Lower mentalizing in parents who maltreat their children. Frontiers in Psychology, 13, 853343. https://doi.org/10.3389/fpsyg.2022.853343
- Rostad, W. L., & Whitaker, D. J. (2016). The association between reflective functioning and parent–child relationship quality. *Journal of Child and Family Studies*, 25(7), 2164–2177. https://doi.org/10.1007/ s10826-016-0388-7
- Salo, S. J., Flykt, M., Mäkelä, J., Biringen, Z., Kalland, M., Pajulo, M., & Punamäki, R. L. (2019). The effectiveness of nurture and play: A mentalisation-based parenting group intervention for prenatally depressed mothers. *Primary Health Care Research & Development*, 20, Article e157. https://doi.org/10. 1017/S1463423619000914
- Sanders, M. R., Morawska, A., Haslam, D. M., Filus, A., & Fletcher, R. (2014). Parenting and family adjustment scales (PAFAS): Validation of a brief parent-report measure for use in assessment of parenting

skills and family relationships. *Child Psychiatry and Human Development*, 45(3), 255–272. https://doi.org/10.1007/s10578-013-0397-3

- Santelices, M.-P., & Cortés, P. A. (2022). Mentalization and parental stress: How do they predict mother–child interactions? *Children*, 9(2), 280. https://doi.org/10.3390/children9020280
- Sharp, C., & Fonagy, P. (2008). The parent's capacity to treat the child as a psychological agent: Constructs, measures and implications for developmental psychopathology. *Social Development*, 17(3), 737–754. https://doi.org/10.1111/j.1467-9507.2007.00457.x
- Sleed, M., Baradon, T., & Fonagy, P. (2013). New beginnings for mothers and babies in prison: A cluster randomized controlled trial. *Attachment & Human Development*, 15(4), 349–367. https://doi.org/10.1080/ 14616734.2013.782651
- Spitzer, R. L., Kroenke, K., Williams, J. B. W., & Löwe, B. (2006). A brief measure for assessing generalized anxiety disorder: The GAD-7. Archives of Internal Medicine, 166(10), 1092–1097. https://doi.org/10. 1001/archinte.166.10.1092
- Suchman, N. E., Decoste, C., Mcmahon, T. J., Rounsaville, B., & Mayes, L. (2011). The mothers and toddlers program, an attachment-based parenting intervention for substance-using women: Results at 6-week follow-up in a randomized clinical pilot. *Infant Mental Health Journal*, 32(4), 427–449. https://doi.org/10. 1002/imhj.20303
- Suchman, N. E., DeCoste, C. L., McMahon, T. J., Dalton, R., Mayes, L. C., & Borelli, J. (2017). Mothering from the inside out: Results of a second randomized clinical trial testing a mentalization-based intervention for mothers in addiction treatment. *Development and Psychopathology*, 29(2), 617–636. https:// doi.org/10.1017/S0954579417000220
- Toth, S. L., & Manly, J. T. (2019). Developmental consequences of child abuse and neglect: Implications for intervention. *Child Development Perspectives*, 13(1), 59–64. https://doi.org/10.1111/cdep.12317
- Weekers, L. C., Hutsebaut, J., & Kamphuis, J. H. (2019). The level of personality functioning scale-brief form 2.0: Update of a brief instrument for assessing level of personality functioning. *Personality and Mental Health*, 13(1), 3–14. https://doi.org/10.1002/pmh.1434
- Woolgar, Matt, Humayun, Sajid, Scott, Stephen, & Dadds, Mark R (2023). I Know What to Do; I Can Do It; It Will Work: The Brief Parental Self Efficacy Scale (BPSES) for Parenting Interventions. *Child Psychiatry Hum Dev*, 1573-3327. https://doi.org/10.1007/s10578-023-01583-0. 37624477.
- Zettl, M., Volkert, J., Vögele, C., Herpertz, S. C., Kubera, K. M., & Taubner, S. (2020). Mentalization and criterion a of the alternative model for personality disorders: Results from a clinical and nonclinical sample. *Personality Disorders*, 11(3), 191–201. https://doi.org/10.1037/per0000356

Author biographies

Lina Gervinskaitė-Paulaitienė PhD is a Researcher of the Developmental Psychopathology Research Center at the Institute of Psychology of Vilnius University, Lithuania and a practicing clinical psychologist.

Matthew Ruggiero is a Consultant Adult and Child Psychologist and Psychotherapist and a Director at Lifespan Psychology Centre, Western Australia.

Svenja Taubner is a Psychoanalyst and Full Professor and Director at the Institute for Psychosocial Prevention at the University of Heidelberg in Germany. Among her many interests is clinical applications, development and research on mentalization based treatments (MBT) and Personality disordered patients. She is an accredited trainer and supervisor of MBT and currently president of the MBT-D-A-CH (MBT association in German-speaking countries).

Jana Volkert is a Professor at the Clinic for Psychosomatic Medicine and Psychotherapy, University of Ulm. Her clinical and research work focuses on transdiagnostic assessments and treatments, as well as mechanisms of change.

Rasa Barkauskienė PhD is a Professor in the Clinical Psychology program and a Head of the Developmental Psychopathology Research Center at the Institute of Psychology of Vilnius University, Lithuania.